Author’s response to reviews

Title: Anti-hyperalgesic properties of a flavanone derivative Poncirin in acute and chronic inflammatory pain models in mice

Authors:

Ruqayya Afridi (r.afridi36@gmail.com)
Ashrafullah Khan (ashrafwazir6@gmail.com)
Sidra Khalid (sidra.merlin@gmail.com)
Busha Shal (bushra.shal@gmail.com)
Hina Rasheed (hinarasheed137@gmail.com)
Zia Ullah (ziaullah121@gmail.com)
Omer Shehzad (omersnu@yahoo.com)
Yeong Shik Kim (kims@snu.ac.kr)
Salman Khan (skhan@qau.edu.pk)

Version: 9 Date: 10 Aug 2019

Author’s response to reviews:

August 10, 2019

BMC-Pharmacology and Toxicology

Editorial Office

RE: Revision PHAT-D-18-00110R8

Dear Editor,

This is the revision eight (R8) of our manuscript for publication in BMC-Pharmacology and Toxicology. We highly appreciate that you potentially accepted our manuscript for publication. Some of the points raised be the editor appended below;
Editor’s comments

1. Please can you ensure to include a statement in your Methods section that clearly states that each experiment uses a different set of mice.

Response:

The statement about the use of animals (each experiment was performed with different set of mice and used once) was added into the method section as per recommendation in the revised manuscript (section “animals”, page 5). However, the changes were not highlighted in the manuscript as per editor recommendation.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response:

The higher education commission of Pakistan only provided financial support in the form of chemicals and equipment and are not involved in the analysis as well as interpretation of the results. However, the Professor Yeong Shik Kim (NRF) despite of the financial support also analyzed the data and assisted in the design as well as interpretation of the results. The changes were made in the section “Funding” page 16. However, the changes were not highlighted in the manuscript as per recommendation.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

Response:

The clean and final version of the manuscript was uploaded with no track changes and comment. The figures and tables were cited within the text according to the recommendation. The figures were uploaded as separate files as per instruction.

We are grateful for your efforts for the improvement of our manuscript and are hopeful for a positive decision this time, on our manuscript.

The final version of the manuscript was edited by the native speaker (www.aje.com).
We would be pleased to consider our manuscript for publication after potentially accepted manuscript in BMC-Pharmacology and Toxicology.

Sincerely yours

Salman Khan (PharmD, Ph.D.)
Assistant Professor of Pharmacology
Department of Pharmacy, Faculty of Biological Sciences, Quaid-i-Azam University, Islamabad, Pakistan
Address: Room No. 03, Department of Pharmacy, Quaid-i-Azam University, Islamabad, Pakistan
Phone: +92-51-90644207
Email: skhan@qau.edu.pk